Author's response to reviews

Title: Could intestinal adhesion due to previous uterine surgery be a risk factor for delayed diagnosis of uterine rupture? A Case report

Authors:
Tomoyuki Kuwata (kuwata@jichi.ac.jp)
Shigeki Matsubara (matsushi@jichi.ac.jp)
Rie Usui (satori@jichi.ac.jp)
Shin-ichiro Uchida (r0806su@jichi.ac.jp)
Naohiro Sata (sata@jichi.ac.jp)
Mitsuaki Suzuki (msuzuki@jichi.ac.jp)

Version: 4 Date: 23 September 2011

Author's response to reviews: see over
Dear Editor

I am pleased to see the favorable comment. I made the following change.

1. “Could intestinal adhesion due to previous uterine surgery be a risk factor for delayed diagnosis of uterine rupture? A Case report.” was employed.

I hope that you would consider that the revised version would be suitable for publication in Journal of Medical Case Reports.

Sincerely,
Tomoyuki Kuwata
Shigeki Matsubara